Author's response to reviews

Title: The Relationship Between Microvascular Complications And Vitamin D Deficiency In Type-2 Diabetes Mellitus

Authors:

Celil Usluogullari (calperuslu@yahoo.com)
Fevzi Balkan (fbalkan21@gmail.com)
Sedat Caner (drsedat@yahoo.com)
rifki ucler (rifkiucler@gmail.com)
Cafer Kaya (caferk@gmail.com)
Reyhan Ersoy (drreyhan@yahoo.com)
Bekir Cakir (drbekir@gmail.com)

Version: 2  Date: 21 August 2014

Author's response to reviews:

Dear editor
We would like to thank the you for your contributions to our manuscript and look forward to hearing from you soon

1-Competing interests
The authors declare that they have no competing interests.

2-Author’s contributions
CAU collected patients, analysis and interpretation of data, and wrote the article, FB made statistics, drafted manuscript, SC participated in the design of the study and performed the statistical analysis, RU collected patients, wrote the article, CF collected patients, RE designed study, coordination and helped to draft the manuscript, BC designed study, coordination and helped to draft the manuscript.

3-We do not have someone to acknowledge to.

4-Our manuscript have 16 page numbers, and 466 line number.